Author's response to reviews

Title: Barriers and facilitators to recruitment of physicians and practices for primary care health services research at one centre

Authors:

Sharon Johnston (sjohnston@bruyere.org)
Clare Liddy (cliddy@bruyere.org)
William Hogg (whogg@uottawa.ca)
Melissa Donskov (mdonskov@bruyere.org)
Grant Russell (grant.russell@med.monash.edu.au)
Elizabeth Gyorfi-Dyke (elizabeth.gyorfidyke@videotron.ca)

Version: 5 Date: 10 December 2010

Author's response to reviews: see over
Editor, BMC Medical Research Methodology

December 10, 2010
Dear Editor,

Re: Manuscript Submission

Thank you for the opportunity to publish our manuscript titled, *Barriers and facilitators to recruitment of physicians and practices for primary care health services research at one centre*, in BMC Medical Research Methodology. This paper compares recruitment strategies, results, and lessons learned from five practice-based primary care research projects in Ontario Canada. It demonstrates that practice and physician recruitment is a crucial yet challenging aspect of primary care research and highlights the need for more quality research to improve our current understanding and methods.

The following authors have contributed to the writing of this manuscript and approved the final version: Drs. Johnston, Liddy, Hogg, and Russell, and Ms. Gyorfi-Dyke and Ms. Donskov.

The final requested major and minor revisions have been made in this the revised manuscript as requested in order to conform to the journal’s formatting style.

I confirm that all authors have seen and agree with the contents of this manuscript and confirm that the work has not been submitted or published elsewhere in whole or in part.

This manuscript is submitted in accordance with the regulations as defined by the BMC Medical Research Methodology editorial policies.

Please do not hesitate to contact me if any further information is required.

Sincerely,

Dr. Sharon Johnston – Corresponding author
Assistant Professor
Department of Family Medicine
University of Ottawa
CT Lamont Primary Health Care Research Centre
Élisabeth Bruyère Research Institute
43 Bruyère Street
Ottawa, On
K1N 5C8
Tel: (613) 562-0050 X 1449